RESEARCH

Journal of Eating Disorders

Open Access

Do risk factors differentiate DSM-5 and drive for thinness severity groups for anorexia nervosa?

An Binh Dang^{1*}, Litza Kiropoulos¹, Marija Anderluh², David Collier³, Fernando Fernandez-Aranda⁴, Andreas Karwautz⁵, Janet Treasure⁶, Gudrun Wagner⁵ and Isabel Krug¹

Abstract

Background The current study examined whether risk factors for anorexia nervosa (AN) were related to different levels of severity based on (a) the DSM-5/body mass index (BMI) and (b) drive for thinness (DT) severity ratings.

Methods The sample comprised 153 pairs of individuals with a lifetime diagnosis AN per DSM-IV criteria and their non-ED sisters (N = 306, mean age = 26.53; mean current BMI = 20.42 kg/m²). The Oxford risk factor interview was used to establish AN-related risk factors. Individuals were categorised into the DSM-5 severity groups based on their lowest BMI, while the DT subscale from the eating disorder inventory-2 was used to classify individuals with AN into low and high DT groups.

Results Multinominal regression models showed similar risk factors (e.g., perfectionism, having a history of being teased about weight and shape) contributed to the development of AN using the DSM-5 and DT severity ratings. Follow-up analyses across the severity groups for both indices revealed that only childhood perfectionism was found to be more common in the extreme severe DSM-5 BMI severity group compared to the severe DSM-5 group.

Conclusion Overall, this study found little evidence for AN risk factors being related to the DSM-5 and DT severity ratings. However, given the novelty of this study, replication of the current results is warranted.

Keywords Anorexia nervosa, Risk factors, Severity rating, Drive for thinness, And body mass index

Plain English summary

Several risk factors, such as childhood obesity, have been found to contribute to the development of Anorexia Nervosa (AN). Yet, we are unsure if there is a set of risk factors that influence different levels of AN severity. While the DSM-5 suggests using BMI to measure severity, recent support favour the usage of drive for thinness (DT) as an alternative severity measure. Therefore, this study aimed to explore risk factors specifically associated with the development of different AN severity levels using both the DSM-5 BMI and DT severity classification systems. We recruited 153 pairs of individuals with a lifetime diagnosis AN per DSM-IV criteria and their non-ED sisters. The Oxford risk factor interview was used to establish AN-related risk factors. We found childhood perfectionism, weight/ shape teasing, childhood obesity, and breast-related embarrassment to be significant risk factors for AN. Additionally, childhood perfectionism was more common in the extreme severe DSM-5 group compared to the severe DSM-5

*Correspondence: An Binh Dang dangb1@student.unimelb.edu.au Full list of author information is available at the end of the article



© The Author(s) 2024. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.gr/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.gr/licenses/by/4.0/. The Creative Commons Public Domain and redit line to the data.

group. This suggests that adding perfectionism-related aspects to prevention and early intervention programs for AN may be beneficial. Considering the novelty of this study, replication of the current results is needed.

Introduction

Risk factors ranging from genetic to psychological factors have been implicated in the aetiology of anorexia nervosa (AN) to inform prevention and treatment programs [1, 2]. To aid the clinical management of AN, the fifth edition of the Diagnostic and Statistical Manual (DSM-5) of Mental Disorders [3] presented a new severity rating for AN, which categorises individuals with AN into the mild, moderate, severe, and extreme severe severity groups based-on body-mass index (BMI). However, studies have found the clinical usefulness of the DSM-5 severity rating for AN to be limited [4, 5]. As such, researchers have explored alternative severity classifications for AN, including transdiagnostic indices (e.g., drive for thinness [DT]) and weight and shape concerns), and found these alternative severity ratings to be superior in indexing eating disorder (ED) psychopathology in comparisons to the DSM-5 BMI severity rating [4]. Despite many studies having been conducted on both AN risk factors [6] and severity ratings [4], researchers have yet to examine risk factors specifically associated with the development of different AN severity levels.

Risk factors for anorexia nervosa

Several risk factors have been implicated in the aetiology of AN, which include amongst others a family history of an ED [7, 8] or traits such as perfectionism, low self-esteem, or obsessive personality traits [8, 9]. However, most of these studies used self-report assessments, and are thus limited by recall biases [10]. The Oxford Risk Factor Interview (ORFI) [11], a semi-structured interview, has been considered the gold standard ED risk factor assessment tool. This is due to the ORFI's ability to assist participants in recalling information related to the autobiographic anchor point and the use of probes to encourage detailed descriptions of risk factor-related events (e.g., parental separation).

Most studies using the ORFI employed case-control designs, which compared individuals with AN to individuals with no DSM-5 ED diagnosis or no psychiatric disorders [6, 12]. These studies [6, 12] mainly found, similar to the studies using self-report measures [8], that perfectionism, negative self-evaluation, weight and shape concerns, and familial history of AN were risk factors associated with the development of AN. However, these ORFI case-control studies are limited by

their inability to control for environmental and family factors that might contribute to the development of AN [13].

To overcome these limitations, studies have started employing samples of sister pairs discordant for AN, i.e., individuals with AN and their non-ED sisters. This design allows for control of cultural, environmental, and family factors so that individual-specific (i.e., non-shared) factors can be more distinctively assessed [14, 15]. Karwautz et al. [14, 15], employing this discordant sister pair design with 90 and 256 participants, respectively, found that individuals with AN, compared to their non-ED sisters, experienced higher rates of weight and shape teasing, exposure to critical comments, and disruptive life events. However, further research is required to elucidate the relationship between varying severity levels of AN and the associated risk factors of AN.

Combining AN risk factors and severity

While the medical literature has extensively investigated the relationships between risk factors and symptom severity in conditions like coronavirus (COVID-19) disease [16] and acute stroke [17], such exploration has been lacking in psychiatric research for conditions such as AN. For example, having a history of obesity and diabetes has been found to increase the risk of having severe COVID-19 (e.g., respiratory failure, intensive care unit admission [16] This means that the medical field has gained a certain understanding of how certain risk factors not only increase the risk of having a disease but also contribute to progressing to a greater severity (e.g., impairments, mortality). Using such information, medical professionals can proactively identify patients at a high risk of developing severe illnesses, which in turn allows them to inform medical decisions such as the need for hospitalisation or treatment frequency. However, in the ED literature, such a severity-risk factor approach has not yet been assessed.

DSM-5 severity rating for AN based on BMI

To date, the most well-known and widely used severity classification system for AN was proposed by the DSM-5 using BMI [4]. This DSM-5 severity rating categorises individuals with AN into "mild" ($\geq 17.0 \text{ kg/m}^2$), "moderate" (16–16.99 kg/m²), "severe" (15–15.99 kg/m²) and "extreme" (<15 kg/m²) severity groups. However, the clinical utility of the BMI specifiers for AN has consistently been questioned by recent research [4, 5]. Dang et al. [4] conducted a meta-analysis (N=22) assessing

all the current DSM-5 severity rating studies for AN, bulimia nervosa (BN) and binge eating disorder (BED). In relation to AN, which comprised five studies [18–22], the review did not find significant differences in ED psychopathology across the DSM-5 BMI severity groups. This means that although BMI has been widely utilised in both clinical practice and empirical research to index severity [23], the level of evidence regarding the DSM-5 AN severity rating is still unknown. If the DSM-5 severity rating is valid, besides indexing the intensity of ED psychopathology, it should also be able to distinguish other clinical information such as the prognosis or factors involved with the development (i.e., risk factors) and maintenance of AN.

Drive for thinness as an alternative severity rating for AN

Given the ambiguity surrounding the DSM-5 AN severity rating, researchers have explored alternative severity classifications for AN such as overvaluation of weight and shape [5, 22], and DT [4, 24]. Studies have consistently found that individuals with AN who scored high on DT had more severe attitudinal and behavioural ED symptoms, as well as more psychiatric comorbidity compared to those with low DT [25, 26]. DT has been considered as another promising alternative severity rating for EDs in general and AN specifically. Krug et al. [24] found that DT provided more significant differences than the DSM-5 BMI severity groups in ED and general psychopathology, with the high DT group scoring higher on these variables than the low DT individuals. Given that previous research has only assessed the clinical utility of DT severity for AN on ED and general psychopathology [4], it is important to expand the scope of severity rating research to look at the relationship between AN-related risk factors and DT as an alternative severity indicator for AN.

The current study

In this study, by using a sister-pair discordant design (comprising individuals with a lifetime DSM-IV diagnosis of AN and a sister without an ED diagnosis), we aimed to identify specific risk factors linked to varying levels of AN severity according to either DSM-5 or DT severity rating. This extends the work of Karwautz et al. [15], who employed a similar design (and part of an overlapping sample) but focused on identifying general risk factors for AN development without delving into specific risk factors contributing to different AN severity levels per DSM-5 and DT ratings. To do so, we first assessed which ORFI risk factors were related to the development of AN based on both the DSM-5 BMI and DT severity ratings, while using non-ED sisters as controls to enhance the distinctiveness of individual factors. We then focused only on the significant risk factors from these primary analyses and examined whether there were significant differences in the frequency of the exposure of each of these significant risk factors among the DSM-5 BMI (mild/ moderate, severe, and extreme severe) and DT (high vs. low) severity groups. Understanding such relationships would not only provide insights into the clinical utility of these two severity classification systems, but would also provide a guide for developing more proactive prevention programs and informing public clinical management policies for AN.

Method

Design

This study used a discordant sister pair design, which meant that women who met the diagnosis of AN according to DSM-IV criteria [27] were compared to their non-ED sisters. Individuals meeting the criteria for AN in the DSM-IV would generally still qualify for an AN diagnosis under the DSM-5 criteria [28]. This is because the DSM-5 broadens the AN criteria, removing specific weight thresholds and the amenorrhea requirement to encompass a wider spectrum of individuals with the disorder. This approach aligns with previous studies in this field [20]. The data for this study were collected as part of a multicentre European project.

Recruitment procedure

Four different sites specialising in EDs participated in this study: the Institute of Psychiatry, London, UK; the University of Vienna, Austria; the University Hospital of Bellvigte, Barcelona, Spain, and the University Children's Hospital, Ljubljana, Slovenia. Between 1999 to 2002, participants were recruited from both clinical settings and community resources such as websites and volunteer databases from previous research studies (for more information please refer to Giles et al. [27].

Participants

The current study comprised 153 pairs of individuals with AN and their non-ED sisters (N=306). Their combined mean age was 26.25 (SD=8.00), and their mean current BMI was 20.42 kg/m² (SD=3.60). Most participants were employed (43.9%). Among the individuals with AN, the mean age of illness onset and duration of illness was 16.98 years (SD=4.92) and 6.51 years (SD=6.59) respectively.

Inclusion criteria for individuals with AN were: (1) female gender; (2) a lifetime diagnosis of AN according to DSM-IV criteria derived from the Longitudinal Interval Follow-up Evaluation [LIFE] interview [29]; and (3) having a sister close in age. The exclusion criteria for both individuals with AN and their non-ED sisters included

not having a current psychotic disorder and learning disability.

Non-ED sisters were included in the study if they did not have a diagnosis of an ED which was screened by the LIFE interview [29]. To control for environmental factors, the sister pairs needed to have an age gap of less than 10 years and must have lived together in the same family for at least 8 years. If the patient had more than one sister, the sister who was closest in age was included.

Measure

Clinical psychologists, with over 5 years of research experience in the field of EDs, conducted all interviews outlined below. These psychologists received training from senior ED clinicians based in different European countries. Although inter-rater reliability was not assessed in this study, prior research [12, 30] has consistently shown adequate reliability and validity for the semi-structured interviews (e.g., LIFE [29], ORFI [11]) employed in the current study.

Social demographic and clinical information

The first part of the EATATE [31], a semi-structured interview to assess ED symptoms, was used to derive information regarding participants' age, occupation, current height, current weight, and highest and lowest weight since reaching adulthood. From this information, the current, lowest, and highest BMI was calculated. In addition, current weight and height were measured at the ED units at intake.

Longitudinal interval follow-up evaluation (LIFE)

This study used an adapted version of the European LIFE [29] to measure lifetime ED history among participants with AN. This included constructing anchor points and timelines for the development of ED symptoms, symptom severity, and psychiatric treatment received. Non-ED sisters were also screened for any ED and ED-related features using the same LIFE interview This was done to ensure that the non-ED sisters have never met any ED diagnoses. This interview has demonstrated adequate reliability and validity in prior research (e.g., 31).

Oxford risk factor interview for eating disorders (ORFI)

The ORFI [11] was used to examine specific risk factors associated with the development of AN. At the beginning of the interview, the clinicians aimed to identify the period before the onset of AN, where the age of onset is defined as the time when the first significant persistent disordered eating pattern began [11, 12, 32]. Drawing from prior research [11, 32], a comprehensive array of potential risk factors was examined and categorised into three overarching domains: individual vulnerability,

environmental risk, and dieting vulnerability. Each domain included several subdomains (e.g., childhood characteristics, childhood abuse), representing specific types of exposure within the main domain. For example, to assess negative self-evaluation, participants were asked, "In general, as a child or adolescent, how did you feel about yourself compared to other people? Did you feel that you were the same, better, or worse than the others?" Another example item to assess the risk of deliberate self-harm included "Have you ever tried to hurt yourself without the intention of committing suicide?". Exposure to a risk factor was rated on a 5-point scale ranging from 0=no exposure to 4=high severity, long duration, or high frequency of exposure. The data were then recoded into 0 = no definite exposure (initially coded 0, 1, or 2) versus 1 = definite exposure to reduce the likelihood of false positives, (initially coded 3 or 4). The ORFI has been found to have good inter-rater reliability with a high level of agreement across the risk factor domains (main weighted kappa: 0.66, SD: \pm 0.17) [12]. For more information, please refer to Fairburn et al. [11].

Eating disorders inventory-2 (EDI-2)

The EDI-2 [33] is a 91-item self-report measure of cognitive and behavioural characteristics frequently associated with EDs. We used the EDI-2 DT subscale as an alternative severity rating for AN. EDI-2 DT subscale includes seven items (rating from 1=never to 6=always), which assesses excessive dieting anxiety, preoccupation with weight, and fear of weight gain (e.g., "I am preoccupied with the desire to lose weight"). The DT subscale has demonstrated excellent internal reliability with α =0.85 [24].

DSM-5 AN severity groups

This study recruited participants with a lifetime diagnosis of AN according to DSM-IV criteria, meaning that some individuals had already recovered from their ED. Therefore, to have an accurate representation of the level of AN severity that the participants had experienced during their illness, we used the lowest lifetime BMI after reaching 18 years old rather than the current BMI to categorise participants into the four BMI AN severity groups.

Using this minimum BMI severity index, we categorised the AN participants according to the four DSM-5 severity groups. However, since there were only a few individuals who were categorised in the mild and moderate groups [*mild* (n = 10), *moderate* (n = 21)], we had to reclassify the mild and the moderate AN DSM-5 severity groups into one "mild/moderate" group. This reclassification has been reported in previous studies [33] to allow a sufficiently large sample size across the groups to undertake meaningful comparisons. Overall, a total of 31 participants were classified as mild/moderate, 40 as severe and 82 as extreme severe using the DSM-5 severity ratings based on the lowest lifetime BMI.

Drive for thinness severity groups

The alternative DT severity rating system was derived from the EDI-2 [33] DT subscale. The DT subscale score was derived by summing the scores of all seven DT items. For screening purposes, Garner [34] recommended a cutoff score of > 14 to differentiate those with high DT from those with low DT, which subsequently has been applied across various research studies (e.g., 24). Using the same cut-off of 14, we classified individuals with AN into low DT (n=60) and high DT (n=69) individuals.

Statistical analysis

Analyses of variance (ANOVA) and chi-square analysis were used to compare sociodemographic and clinically related information between the non-ED sisters and the two AN severity indices (DSM-5 BMI and DT) for continuous and categorical variables respectively. Etasquared coefficient (η^2) as the measurement of effect size for ANOVA (values of 0.06, 0.10, and 0.25 were interpreted as low-poor, moderate-medium, and large-high effect sizes respectively; 36). Cramer's V coefficient was used as a measurement of effect size for chi-square analyses (values of 0.06; 0.15, and 0.30 were interpreted as low-poor, moderate-medium, and large-high effect sizes respectively [35]). Each risk factor was coded as 0 for "no – the risk was not presented", and 1 for "yes – the risk was present".

Multinominal logistic regression analyses with "non-ED sisters" as the reference group were employed to investigate which risk factors were significantly associated with higher odds of developing the different AN severity categories based on the DSM-5 BMI and DT severity ratings. To account for multiple testing, we applied the Benjamini and Hochberg correction for multiple comparison method, where *p*-values were ranked, and then adjusted by multiplying each by a factor of m/k, with m representing the number of independent tests and k being the position of a *p*-value in the sorted array [36]. The chosen alpha level was 0.05. The results of the significant tests (*p*-value) and the effect-size estimates were utilised to evaluate the validity of both the DSM-5 BMI and the alternative DT severity ratings for AN.

The odds ratio (OR) was used as the measurement of the effects of being exposed to AN-related risk factors and the development of different levels of AN severity based on the DSM-5 BMI and DT severity ratings. An OR of greater than 1.00 indicates that exposure was associated with higher odds of developing AN. With an OR of less than 1.0 is associated with lower odds of developing AN. Following Chen's [37] rules-of-thumb OR of 1.68, 3.47, and 6.71 are considered as small, medium, and large effect-size respectively. Chi-square tests were conducted to test for the differences in the rate of being exposed to each significant risk factor for the DSM-5 BMI and DT severity groups.

Result

Sociodemographic and clinical-related information

The sociodemographic and clinical-related information is displayed in Table 1. As expected, there were significant differences in lowest (p < 0.001), current (p < 0.001), and highest (p < 0.001) BMI between individuals with AN and their non-ED sisters. There were no significant differences in age between non-ED sisters and individuals with AN. Further information on sociodemographic of the sample can be found in Additional file 1: Table S1.

Using the DSM-5 BMI severity rating for AN, significant differences between the BMI severity groups were found for the age of illness onset (p=0.029). However, post-hoc comparisons did not reveal where the differences between the groups lay (Table 1). No other significant differences emerged in the sociodemographic and clinically related information between the DSM-5 BMI severity groups. For the DT groups, no significant differences in any of the sociodemographic (Additional file 1: Table S1) and clinically related information were observed.

Exposure to AN-related risk factors and odds of being in the DSM-5 severity groups

Using non-ED sisters as a reference group, multinomial logistic regression models (Table 2) showed that the following factors increase the risk of developing an AN diagnosis across the DSM-5 BMI severity groups: childhood perfectionism (mild/moderate, OR=6.67; severe, OR=3.13; extreme severe, OR=10.00); childhood obesity (mild/moderate, OR=25.00; severe, OR=20.00; extreme severe, OR=16.67); having a history of being teased about weight, shape, or appearance (mild/moderate, OR=3.86); and having a history of feeling embarrassment about one's breasts (mild/moderate, OR=7.14; severe, OR=5.00; extreme severe, OR=4.35).

Follow-up chi-square analyses were conducted on the risk factors that significantly contributed to the development of AN based on the DSM-5 BMI severity classification (Table 3). Among all the significant risk factors, significant differences between the DSM-5 BMI severity groups were only obtained for childhood perfectionism (p=0.039), with individuals in the extreme severe DSM-5 BMI severity group presenting with more childhood perfectionism compared to individuals in the severe DSM-5

		Comparisons be individuals with	etween non-ED s i AN	sisters and	Comparisons be	tween DSM-!	5 severity group	S	Comparison thinness (DT	s between D) severity gr	rive for oups
	Total Mean (SD)	Non-ED sisters	Individuals with AN	<i>P</i> values (η ²)	Mild/Moderate	Severe	Extreme Severe	<i>P</i> values (η ²)	Low DT	High DT	<i>P</i> values (η²)
Clinical informa- tion											
Age (yrs-old)	26.25 (8.00)	26.55 (8.30)	25.97 (7.73)	0.516 (0.001)	26.72 (6.67)	25.76 (6.05)	25.78 (8.81)	0.826 (0.003)	25.19 (6.72)	26.30 (7.78)	0.156 (0.008)
Onset ED (yrs- old)	16.98 (4.92)	I	16.98 (4.92)	I	17.71 (5.01)	17.41 (4.18)	15.82 (3.36)	0.029 (0.046)	16.52 (4.30)	16.50 (3.82)	0.713 (0.001)
Duration ED (yrs)	6.51 (6.59)	I	6.51 (6.59)	I	6.21 (6.86)	5.36 (3.97)	7.21 (7.11)	0.314 (0.015)	6.13 (6.07)	6.93 (6.11)	0.465 (0.004)
Current BMI (kg/ m ²)	20.42 (3.60)	22.08 (3.90)	18.74 (2.41)	<0.001 (0.217)	19.30 (1.52)	19.30 (2.30)	18.31 (2.68)	0.063 (0.036)	18.63 (2.34)	19.07 (2.47)	0.078 (0.013)
Lowest BMI (kg/ m ²)	17.11 (3.61)	19.81 (2.82)	14.48 (2.00)	<0.001 (0.547)	17.04 (1.17)	15.45 (0.30)	13.04 (1.36)	<0.001 (0.734)	14.47 (2.18)	14.50 (1.81)	0.933 (0.001)
Highest BMI (kg/ m ²)	22.58 (4.19)	23.58 (4.78)	21.56 (3.20)	<0.001 (0.059)	22.22 (2.80)	21.17 (3.65)	21.02 (3.03)	0.082 (0.033)	21.12 (3.18)	21.62 (3.18)	0.075 (0.013)
Social informa- tion				P values (V)				P values (V)			P values (V)
Bold significant con	nparison using adju:	sted p-value using th	ie Benjamini-Hochk	oerg false discovery ra	te procedure (.05 lev	/el)					

 Table 1
 Social and clinical related information of non-ED sisters and individuals with AN

thinness groups	
and low drive for	
groups, and high	
3M-5/BMI severity	
me severe AN DS	
ere and the extre	eference group
noderate, the sev	ED sisters as the n
across the mild/r	ession, with non-
of risk factors for	ninal logistic regr
2 Distribution	s using multinor.
Table	result

Severity groups	DSM-5 Mild/Mode	erate	DSM-5 Severe		DSM-5 Extreme	Low DT		High DT	
	OR [95% CI]	р	OR [95% CI]	р	OR [95% CI] µ	0R [95% CI]	d	OR [95% CI]	d
Individual vulnerability domain									
Subdomain 1 (childhood characteristic)									
Negative self-evaluation	0.58 [0.26; 1.35]	0.455	1.15 [0.55; 2.44]	0.885	1.10 [0.59; 2.04]	0.890 0.62 [0.31; 1.22] 0.415	1.69 [0.84; 3.45]	0.391
Perfectionism	6.67 [2.33; 20.00]	0.006	3.13 [1.20; 8.33]	0.003	10.00 [4.55; 20.00]	< 0.001 8.26 [3.57; 19.2	3] <0.001	I 6.67 [2.86; 14.29]	< 0.0001
Shyness	1.96 [0.83; 4.76]	0.372	1.15 [0.53; 2.50]	0.906	0.91 [0.47; 1.79]	0.897 0.90 [0.43; 1.87] 0.894	1.39 [0.69; 2.86]	0.631
Extreme compliance	1.05 [0.42; 2.63]	0.952	1.89 [0.87; 4.00]	0.366	1.52 [0.79; 2.86]	0.451 1.27 [0.62; 2.60	0.771	1.75 [0.87; 3.57]	0.359
No close friends	0.42 [0.13; 1.37]	0.407	1.01 [0.43; 2.38]	0.978	0.46 [0.21; 1.02]	0.232 1.03 [0.46; 2.32] 0.951	0.38 [0.15; 0.93]	0.169
No opposite gender friends	0.43 [0.15; 1.25]	0.365	0.55 [0.23; 1.30]	0.416	1.10 [0.55; 2.17]	0.895 0.67 [0.30; 1.45	0.610	0.78 [0.36; 1.69]	0.773
School absence through anxiety	I	I	0.11 [0.01; 0.88]	0.178	0.14 [0.03; 0.66]	0.089 -	I	0.27 [0.07; 1.00]	0.216
Subdomain 2 (premorbid psychiatric disorder)									
Major depression	1.89 [0.75; 4.76]	0.419	2.94 [1.35; 6.67]	0.051	2.00 [1.05; 3.85]	0.180 2.04 [0.97; 4.17	0.232	2.50 [1.22; 5.00]	0.087
Anxiety disorders	1.23 [0.51; 3.03]	0.839	2.04 [0.96; 4.35]	0.239	1.47 [0.80; 2.63]	0.444 1.19 [0.60; 2.38	0.827	1.96 [1.01; 3.85]	0.207
Substance abuse	0.65 [0.07; 6.25]	0.889	0.48 [0.55; 4.55]	0.774	1.23 [0.31; 5.00]	0.894 0.74 [0.07; 7.14	0.883	3.33 [0.75; 14.29]	0.361
Soft drug abuse	3.57 [1.43; 9.09]	0.057	2.13 [0.85; 5.26]	0.368	1.23 [0.55; 2.78]	0.827 0.88 [0.34; 2.27	0.884	2.13 [0.94; 4.76]	0.249
Subdomain 3 (behavioral problems)									
Marked conduct problem	1.11 [0.80; 100]	0.263	2.94 [0.15; 50.00]	0.734	1.59 [0.09; 25.00]	0.893 -	I	5.56 [0.49; 50.00]	0.411
School absence (truancy)	0.72 [0.10; 5.26]	0.889	1.41 [0.32; 6.25]	0.846	1.23 [0.38; 4.00]	0.897 0.64 [0.13; 3.13	0.813	1.67 [0.51; 5.56]	0.678
Deliberate self-harm	7.14 [1.75; 25.00]	0.053	5.56 [1.39; 20.00]	0.097	4.17 [1.22; 14.29]	0.127 4.17 [0.97; 16.6	7] 0.233	8.33 [2.08; 33.33]	0:030
Subdomain 4 (physical health)									
Severe personal health problems	2.22 [0.83; 5.88]	0.363	1.37 [0.51; 3.57]	0.781	1.33 [0.62; 2.94]	0.718 1.56 [0.68; 3.70	0.558	2.00 [0.85; 4.55]	0.362
Childhood obesity	25.00 [7.14; 100.00]	< 0.001	20.00 [5.88; 100.00]	< 0.001	16.67 [4.76; 50.00]	< 0.001 25.00 [5.26; 10	0.00] 0.001	33.33 [7.69; 100.00]	< 0.001
Subdomain 5 (parental psychiatric disorder)									
Parental depression	1.14 [0.40; 3.23]	0.877	0.68 [0.23; 2.00]	0.733	1.38 [0.68; 2.78]	0.650 1.82 [0.83; 4.00	0.377	0.82 [0.34; 2.00]	0.849
Parental anxiety	0.56 [0.06; 4.76]	0.811	2.38 [0.65; 8.33]	0.435	2.08 [0.75; 5.88]	0.419 0.82 [0.20; 3.45	0.891	2.33 [0.70; 7.69]	0.418
Parental drug abuse	1.69 [0.55; 5.26]	0.635	0.75 [0.20; 2.86]	0.853	1.03 [0.43; 2.50]	0.961 0.71 [0.25; 2.00	0.775	1.23 [0.47; 3.23]	0.848
Environmental Risk Domain									
Subdomain 1 (parental problem)									
Distress due to low parental contact	0.76 [0.15; 3.85]	0.892	1.18 [0.35; 3.85]	0.886	1.82 [0.72; 4.55]	0.462 0.65 [0.19; 2.27	0.749	1.55 [0.58; 4.15]	0.651
Distress due to separation from parents	0.84 [0.16; 4.55]	0.890	0.59 [0.11; 3.23]	0.782	0.43 [0.12; 1.56]	0.447 0.35 [0.07; 1.81	0.442	0.57 [0.14; 2.36]	0.713
Distress due to parental arguments	2.13 [0.92; 5.00]	0.284	0.99 [0.44; 2.22]	0.984	2.63 [1.47; 4.76]	0.142 3.31 [1.70; 6.45	0.006	1.08 [0.55; 2.12]	0.877
Parental criticisms	1.54 [0.55; 4.35]	0.682	1.03 [0.38; 2.78]	0.968	1.45 [0.69; 3.03]	0.609 1.11 [0.47; 2.60	0.887	1.56 [0.69; 3.56]	0.558
Parental high expectation	1.39 [0.59; 3.23]	0.724	0.74 [0.33; 1.64]	0.726	1.53 [0.85; 2.78]	0.419 1.89 [0.97 3.70]	0.239	0.92 [0.46; 1.82]	0.885

(continued)
2
ľq
Ē

Severity groups	DSM-5 Mild/Mod	erate	DSM-5 Severe		DSM-5 Extreme	Low DT		High DT	
	OR [95% CI]	р	OR [95% CI]	þ	OR [95% CI]	p OR [95% CI]	р	OR [95% CI]	р
Parental over-involvement	0.88 [0.33; 2.38]	0.880	1.18 [0.46; 3.03]	0.902	0.84 [0.41; 1.72]	0.837 0.74 [0.32; 1.70]	0.731	1.21 [0.56; 2.63]	0.836
Parental neglect	0.31 [0.03; 2.86]	0.564	4.35 [1.37; 14.29]	0.092	0.56 [0.17; 1.92]	0.639 0.69 [0.18; 2.60]	0.813	1.77 [0.63; 5.00]	0.547
Parental affection	0.75 [0.29; 1.96]	062.0	1.39 [0.58; 3.33]	0.719	0.70 [0.35; 1.37]	0.528 0.76 [0.36; 1.62]	0.479	1.32 [0.63; 2.75]	0.730
Subdomain 2 (disruptive events)									
Distress due to parental death	3.33 [0.51; 20.00]	0.451	2.00 [0.30; 12.50]	0.733	1.27 [0.20; 7.69]	0.886 0.73 [0.07; 7.41]	0.889	3.68 [0.81; 16.67]	0.314
Distress due to parental chronic illness	1.20 [0.35; 4.00]	0.894	2.13 [0,81; 5.56]	0.375	0.95 [0.38; 2.44]	0.947 1.17 [0.44; 3.11]	0.892	1.51 [0.60; 3.77]	0.655
Distress due to frequent house moves	2.00 [0.37; 11.11]	0.691	3.13 [0.78; 12.50]	0.364	3.33 [1.04; 10.00]	0.193 2.47 [0.68; 8.93]	0.422	2.29 [0.62; 8.47]	0.444
Subdomain 3 (teasing and bullying)									
Teasing (not concerning weight, shape, eating or appearance)	0.84 [0.23; 3.06]	0.894	1.37 [0.50; 3.75]	0.779	2.29 [1.12; 4.67]	0.123 1.27 [0.53; 3.04]	0.815	1.88 [0.83; 4.27]	0.372
Bullying	0.24 [0.05; 1.04]	0.237	0.27 [0.08; 0.92]	0.181	0.38 [0.17; 0.85]	0.113 0.35 [0.14; 0.89]	0.147	0.22 [0.07; 0.66]	0.055
Subdomain 4 (sexual and physical abuse)									
Sexual abuse	1.31 [0.53; 3.25]	0.795	2.39 [1.13; 5.08]	0.130	1.34 [0.72; 2.51]	0.633 1.41 [0.70; 2.82]	0.618	1.36 [0.68; 2.72]	0.653
Physical abuse	0.57 [0.18; 1.79]	0.614	1.22 [0.53; 2.82]	0.842	0.92 [0.46; 1.84]	0.883 0.72 [0.32; 1.63]	0.707	1.35 [0.65; 2.79]	0.686
Dieting Vulnerability Domain									
Subdomain 1 (dieting risk)									
Influenced by parent eating disorders	1.63 [0.27; 9.71]	0.810	1.35 [0.24; 7.69]	0.890	0.30 [0.03; 2.81]	0.562 1.13 [0.23; 5.46]	0.924	0.84 [0.15; 4.69]	0.889
Family member dieting for shape and weight	1.96 [0.76; 5.03]	0.419	1.05 [0.43; 2.58]	0.953	0.89 [0.42; 1.87]	0.895 0.91 [0.40; 2.07]	0.873	1.15 [0.53; 2.50]	0.893
Critical comments by family about weight, shape or eating	1.05 [0.45; 2.46]	0.947	1.77 [0.84; 3.76]	0.373	1.19 [0.65; 2.15]	0.811 1.16 [0.59; 2.27]	0.857	1.35 [0.70; 2.62]	0.644
Repeated comments by others about shape and weight	0.75 [0.24; 2.33]	0.832	1.74 [0.73; 4.17]	0.446	1.56 [0.76; 3.19]	0.462 1.18 [0.51; 2.74]	0.886	1.61 [0.73; 3.53]	0.472
Teasing about shape, weight, or appear- ance	4.50 [1.86; 10.99]	0.010	3.40 [1.48; 7.81]	0.038	3.86 [1.95; 7.63]	0.002 4.93 [2.40; 10.10]	< 0.001	3.61 [1.75; 7.46]	0.007
Dieting at school common	2.48 [0.76; 8.06]	0.379	2.20 [0.73; 6.58]	0.416	2.04 [0.83; 5.03]	0.361 2.23 [0.83; 5.95]	0.367	1.82 [0.66; 5.05]	0.495
Subdomain 2 (pudicity domain)									
Preparation for menarche	0.45 [0.14; 1.43]	0.421	0.86 [0.36; 2.04]	0.895	0.64 [0.31; 1.33]	0.476 0.96 [0.43; 2.15]	0.950	0.57 [0.24; 1.37]	0.453
Menarche mainly unpleasant	2.86 [0.96; 8.33]	0.236	2.04 [0.74; 5.88]	0.416	3.03 [1.32; 6.67]	0.066 3.32 [1.22; 9.01]	0.110	4.90 [1.96; 12.35]	0.008
Scared by early breast development	0.29 [0.09; 0.93]	0.182	0.56 [0.23; 1.35]	0.449	0.41 [0.19; 0.85]	0.113 0.30 [0.13; 0.73]	0.061	0.39 [0.17; 0.89]	0.458
Teasing about breast as early age	0.32 [0.10; 1.09]	0.254	0.51 [0.19; 1.32]	0.417	0.22 [0.09; 0.57]	0.021 -	I	0.40 [0.15; 1.01]	0.233
Breast as source of embarrassment	7.14 [2.78; 16.67]	0.001	5.00 [2.08; 11.11]	0.004	4.35 [2.13; 9.09]	0.001 3.39 [1.48; 7.75]	0.037	5.41 [2.48; 11.76]	0.001
Adjusted p-value using the Benjamini-Hochberc	g false discovery rate pi	rocedure							

	DSM-	Ń							Б				
	Mild/ mod€	erate (1)	Sevel	'e (2)	Extre	me (3)	χ ² (<i>p</i> value; Cramer's V)	Post-hoc	Low		High		χ ² (<i>p</i> value; Cramer's V)
	z	%	2	%	2	%			2	%	2	%	
Perfectionism	11	17.2	1	17.2	42	65.6	6.86 (0.032; 0.212)	3 > 2 (.039)	31	47	35	53	0.43 (0.515; 0.042)
Childhood obesity	1	25.6	12	27.9	20	46.5	1.45 (0.480; 0.098)	I	15	40.5	22	59.5	1.70 (0.193; 0.083)
Teasing about shape, weight, or appearance	13	23.2	14	25	29	51.8	0.48 (0.787; 0.056)	I	34	50	34	50	0.01 (0.937; 0.005)
Breast as source of embarrassment	4	15.4	6	34.6	13	50	1.30 (0.521; 0.092)	I	22	39.3	34	60.7	3.59 (0.058; 0.121)

Table 3 Follow-up chi-square analyses on the risk factors that significantly contributed to the development of AN based on the DSM/BT and DT classification

BMI severity group. No other significant differences between the DSM-5 BMI severity groups in the rate of being exposed to the remaining significant AN risk factors were found.

Exposure to AN-related risk factors and odds of being in the high/low DT severity groups

For DT, multinomial logistic regression models (using non-ED sisters as the comparison group to either low or high DT) found childhood perfectionism (low DT, OR=8.26; high DT, OR=6.67); childhood obesity (low DT, OR = 25.00; high DT, OR = 33.33); having a history of being teased about weight, shape, or appearance (low DT, OR = 4.93; high DT, OR = 3.61); and feeling embarrassed about breasts (low DT, OR = 3.39; high DT, OR = 5.41) to be significant risk factors for both the low and high DT groups. Conversely, in comparison to non-ED sisters, having a history of self-harm was found to only increase the risk of being assigned to the high DT AN severity group (OR=8.33), but not the low DT group. Additionally, in comparison to non-ED sisters, those in the low DT group were more likely to feel distress due to parental arguments (OR = 3.31; Table 2).

Follow-up chi-square analyses were conducted on the risk factors that significantly contributed to the development of AN based on the DT classification (Table 3). However, these analyses showed no significant differences between the low and high DT groups in the frequency of exposure to each of these risk factors.

Discussion

Using a discordant AN sister pair design, the current study identified risk factors that may contribute to the development of different AN severity levels based on the DSM-5 BMI and DT severity classification systems. Several factors (e.g., perfectionism, teasing about weight and shape – see details below) were found to contribute to the development of AN based on both severity indices. However, when looking at the differences in the rate of being exposed to these significant risk factors across the DSM-5 BMI and DT severity groups, besides childhood perfectionism, no other significant risk factor was found.

Risk factors for AN based on the DSM-5 and DT severity indices

Comparisons between non-ED sisters and AN DSM-5 and DT severity groups

Compared to non-ED sisters, we found that almost the same AN risk factors were associated with the severity spectrum of the DSM-5 BMI and DT classification systems. In accordance with the literature, we found that childhood obesity [38, 39], perfectionism [39, 40]; being teased about weight and shape [41] and feeling embarrassed about breasts [42] were associated with having a subsequent diagnosis of AN using these two severity indices.

This overlap in AN risk factors across the DSM-5 BMI and DT severity indices may imply that these two severity classification systems could be correlated with each other. However, in the current study, opposite patterns in the strength of effects across the two severity ratings were found for several risk factors (e.g., childhood obesity). For example, in comparison to non-ED sisters, childhood obesity was associated with a higher risk of being classified into the mild/moderate DSM-5 BMI severity groups. Contrastingly, those reported having childhood obesity were found to have a higher likelihood of being classified into the high DT (as opposed to low DT) severity group.

These patterns of results are somewhat consistent with the literature [43, 44]. Specifically, studies have shown that a higher childhood BMI was directly associated with an elevated adulthood BMI (i.e., mild/moderate DSM-5 BMI severity) and indirectly through both DT in childhood and adulthood (high DT group; 44). This may suggest that different underlying mechanisms might mediate the relationships between AN risk factors and AN severity levels based on these two severity classifications. Further research using longitudinal designs is needed to establish a temporal relationship of the potential different mechanisms that might lead to the different DSM-5 BMI and DT severity groups when compared to a non-ED control sample.

Differences across DSM-5 and DT severity groups in the significant risk factors

Across all DSM-5 and DT severity groups, those with an AN diagnosis were more likely to exhibit childhood perfectionism. However, when assessing differences across the severity groups, only those in the extreme DSM-5 BMI severity group were significantly more likely to present with childhood perfectionism compared to those in the DSM-5 severe group. This finding aligns with previous studies [45, 46], which found that comorbid perfectionistic tendencies in adulthood were higher in AN individuals with a lower BMI (i.e., severe AN group) than those with a higher BMI (i.e., mild severity AN group). Contrastingly, a recent study by Krug et al. [24] did not find significant differences in adult clinical perfectionism across both the DSM-5 BMI and DT severity groups among individuals with AN. These discrepancies in findings may be either due to the heterogeneous samples across these studies or may have been confounded by the level of care obtained by individuals across these studies. For instance, Dakanalis et al. [45] assessed an inpatient clinical AN sample, while both Krug et al. [24] and the current study comprised a mixture of inpatient,

strongly associated with adulthood perfectionism [48]. No significant differences between either the DSM-5 BMI or DT AN severity groups in the rates of being exposed to the remaining significant risk factors were revealed in our follow-up analyses. Of interest, no significant differences in the frequency of childhood obesity across both the DSM-5 BMI and DT AN severity groups were found. Such findings are unexpected because in the current study childhood obesity was the strongest significant predictor for a subsequent AN diagnosis (OR ranges from 17 to 25). Furthermore, studies have consistently found that individuals with AN with premorbid obesity presented with significantly more severe physical (e.g., cardiovascular failure) and psychological sequelae, and poorer prognosis compared to AN individuals with premorbid normal weight [49, 50]. The current non-significant findings might be attributable to a small sample size across the three DSM-5 BMI and the two DT groups, lacking sufficient power to detect differences across severity groups.

Only Dang et al. [4] and Krug et al. [24] have compared the DSM-5 with the DT severity ratings for all EDs. Findings from both studies support the clinical utility of the DT severity rating over the DSM-5 in indexing ED and general psychopathology across most ED subtypes. In the current study, our regression analyses, however, did not reveal any significant association between the evaluated risk factors and the severity of AN, as determined by either the DSM-5 BMI or DT ratings. Previous studies [5, 22, 24] examined the clinical validity (i.e., significant differences in the level of symptomatology across severity groups), whereas the current study focused on the retrospective validity (i.e., ability to distinguish risk factors that are related to the development of different AN severity levels) of these severity ratings. Therefore, the DT severity rating for AN might only be superior to the DSM-5 BMI severity rating in distinguishing the level of AN-related concurrent symptomatology [4, 24], but may not be as effective in capturing retrospectively assessed risk factors. Nevertheless, this study was the first to examine the relationship between the DSM-5 BMI and DT severity indices and AN risk factors, hence replication of these results is needed.

Implications

The staging model by Treasure et al. [51] can be applied to develop prevention and early treatment programs for AN. Based on our findings such an approach should use two steps. First, it would be important to identify individuals at risk of developing AN where significant AN risk factors (i.e., as per the current results, childhood obesity, being teased about weight, shape, and appearance, and feeling embarrassed about breasts) occur. It might be beneficial for these risk factors to be included in prevention programs for AN across all severity spectrums. As shown to be beneficial in previous research [52], cognitive dissonance-based interventions could aim to increase individuals' appreciation of different body types, reducing pressure to be thin, and body dissatisfaction.

Second, we found that individuals with AN who exhibited perfectionistic traits during childhood were more likely to develop extreme severe AN per the DSM-5 BMI severity rating in adulthood. Such findings, therefore, underscore the importance of incorporating perfectionism in early prevention programs to prevent this trait from intensifying throughout adolescence and adulthood. While it might be advantageous to focus on perfectionism across all AN severity levels (e.g., identifying perfectionism triggers), implementing more comprehensive and targeted early perfectionism interventions (e.g., cognitive remediation therapy [CRT]; [53] might be highly beneficial for individuals who are at risk of developing extreme severe AN. Accordingly, CRT has demonstrated promising results in improving ED symptomatology among individuals with severe and enduring AN [54, 55]. Given the novelty of our current results, future research is warranted to explore the efficacy of additional perfectionism interventions, including CRT modules, in mitigating the progression of AN severity.

Finally, except for perfectionism, we found no other risk factors linked to varying DSM-5 AN severity levels, once again casting doubt on the validity of the proposed BMI-based severity rating by the DSM-5. While recognising the importance of severity ratings in prevention strategies and treatment planning, relying solely on BMI may oversimplify the complex nature of AN, encompassing elements such as fear of weight gain and body distortions [56]. This critique does not dismiss the use of a severity index for AN but underscores the necessity for an alternative perspective based on empirical data. Subsequent research should explore different indicators, such as duration of illness or perfectionism, to refine severity assessments, hence enhancing the efficacy of prevention and treatment strategies. Notably, shorter durations of illness have been proposed as a valuable severity indicator for AN [57, 58], correlating with higher BMI [57], more favourable treatment outcomes [58, 59], lower levels of perfectionism [60], and increased cognitive flexibility [61].

Strengths and limitations

A strength of this research was our discordant case–control interview-based design to identify a wide range of psychosocial risk factors for AN. This means the current study was able to control for environmental factors which allowed to establish individual risk factors.

The current study also has a few limitations that need to be acknowledged. First, along with the exploratory nature of our research, we needed to categorise our difficult-to-recruit 153 AN sample into three DSM-5 BMI and two DT groups. This may not have provided sufficient power to detect a significant relationship between the assessed risk factors and the different DSM-5 severity levels. Future studies should employ a larger sample to ensure a more balanced distribution across the different DMS-5 BMI severity groups to gain a more accurate insight into the relationship between AN severity groups and risk factors.

Second, despite using the ORFI [11] to maximise reporting accuracy, potential recall biases are inherent to retrospective reporting. We did not include other informants in assisting with reporting family history but rather solely relying on the family history stated by the participants. Future research should include family members in the study process to bypass the inherent recall bias of retrospective assessment. Future studies should also consider a longitudinal design to assess the temporal relationship of the ORFI risk factors using different informants.

Third, our study departed from DSM-5 guidelines, categorising AN severity based on lowest BMI instead of current BMI. The lowest BMI was utilised because we recruited participants with a lifetime AN diagnosis, potentially including recovered and non-recovered cases. The inclusion of both groups poses a confounding factor, as non-recovered individuals might reach lower BMIs, leading to shifts between DSM-5 severity groups. Furthermore, the divergence between the DSM-5 severity groups based on lowest BMI and our DT severity groups, which were based on DT levels at the time of assessment, alongside with having a mixture of inpatient, outpatient, and community-based individuals with AN recruited across Europe might have confounded the results of the current study. Therefore, future studies should use the current BMI as per the DSM-5, instead of the lowest BMI, to examine the relationship between AN severity groups and risk factors in a sample of individuals with a current DSM-5 diagnosis of AN.

Fourth, our study exclusively included a female sample, while research has shown that different risk factors contribute to the development of AN in males and females [62]. As a result, the findings of the current study cannot be generalised beyond female-specific sample.

Lastly, in highlighting childhood perfectionism as a significant risk factor for extreme severe AN based on the DSM-5 severity rating, it is crucial to acknowledge a potential limitation in the scope of this finding. The DSM-5 AN severity level can shift from DSM-5 mild to DSM-5 extreme with just a one-point change in BMI. The DSM-5 extreme category, on the other hand, has a broad range (e.g., our participants' BMI in the extreme group had BMI ranged from 8.6 to 15 kg/m²), highlighting potential heterogeneity within the DSM-5 extreme severe group that requires a more detailed exploration for certainty about our study's implications. Future research should specify the DSM-5 extreme severe group into subgroups (e.g., very extreme severe DSM-5 group) for a detailed understanding of diverse characteristics and implications within the extreme severity spectrum.

Conclusion

Our findings provided limited support for the DSM-5 BMI and DT AN severity ratings in identifying related risk factors for varying AN severity levels. However, BMI is an objective physical severity measure whereas DT is a cognitive measure of severity. In treatment, weight changes rapidly (especially in youth), but cognitive symptoms linger, highlighting the need to consider their distinct characteristics within a treatment context. Regardless, the current findings do highlight the significance of childhood perfectionism as a risk factor in developing extreme severe AN according to the DSM-5 BMI severity rating. It may be beneficial for prevention and early intervention programs to include a focus on managing perfectionism to reduce the severity of AN. As our study is the first of its kind, further replication, using several informants and longitudinal designs is needed to provide further insight into the relationship between risk factors for AN and DSM-5 BMI and other alternative severity ratings systems, including weight and shape concerns and, the number of purging methods.

Abbreviations

Abbrevia	tions
AN	Anorexia nervosa
ANOVA	Analyses of variance
BED	Binge eating disorder
BMI	Body mass index
BN	Bulimia nervosa
CRT	Cognitive remediation therapy
DSM	Diagnostic statistical manual
DT	Drive for thinness
ED	Eating disorder
EDI-2	Eating disorder inventory-2
LIFE	Longitudinal interval follow-up evaluation
OR	Odd ratio
ORFI	Oxford risk factor interview

Supplementary Information

The online version contains supplementary material available at https://doi.org/10.1186/s40337-024-00966-5.

Additional file 1. Social and clinical related information of healthy sisters and individuals with AN.

Acknowledgements

Not applicable.

Author contributions

AD: Conceptualization, Methodology, formal analyses, investigation, writing original draft. LK: Writing—review and editing, supervision. MA, DC, FFA, AK, JT and GW: Data collection, writing—review and editing. IK: Methodology, writing—review and editing, supervision, project administration.

Funding

Financial support was received from the European Union (Framework-V Multicenter Research Grant, QCK1-1999-916), Fondo de Investigacion Sanitaria (CIBER; CB06/03/0034) and Generalitat de Catalunya (2005SGR00322). The CIBER Fisiopatologia de la Obesidad y Nutricion is an initiative of ISCIII. All these funding sources had no further role in study design; in the collection, analysis, and interpretation of data; in the writing of the report; or in the decision to submit the paper for publication.

Availability of data and materials

All data analysed during this study are included in this article (tables/figures). Raw data can be requested on reasonable requests.

Declarations

Ethical approval and consent to participate

Ethical approvals were obtained from all participated hospitals and research organisations. Consents were obtained from all participants in accordance with the Declaration of Helsinki.

Consent for publication

Not applicable.

Competing interests

The authors declare that they have no competing interests.

Author details

¹ Melbourne School of Psychological Sciences, The University of Melbourne, Redmond Barry Building, Level 7, Room 707, Melbourne, VIC, Australia. ²University Children's Hospital, University Medical Center Ljubljana SI, Ljubljana, Slovenia. ³ Eating Disorders Unit and SGDP Research Centre, Institute of Psychiatry, King's College, London, UK. ⁴Department of Psychiatry, University Hospital of Bellvitge- IDIBELL, Barcelona, Spain. ⁵Department of Child and Adolescent Psychiatry, Medical University of Vienna, Vienna, Austria. ⁶Section of Eating Disorders, Department of Psychological Medicine, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK.

Received: 19 August 2023 Accepted: 5 January 2024 Published online: 11 January 2024

References

- Jacobi C. Psychosocial risk factors for eating disorders. Eat Disord Rev. 2005;1:59–84.
- 2. Stice E, Whitenton K. Risk factors for body dissatisfaction in adolescent girls: a longitudinal investigation. Dev Psychol. 2002;38(5):669.
- American Psychiatric Association D, Association AP. Diagnostic and statistical manual of mental disorders: DSM-5. Washington, DC: American Psychiatric Association; 2013.

- Dang AB, Giles S, Fuller-Tyszkiewicz M, Kiropoulos L, Krug I. A systematic review and meta-analysis on the DSM-5 severity ratings for eating disorders. Clin Psychol Sci Pract. 2022;29(4):325–44.
- Dang AB, Kiropoulos L, Castle DJ, Jenkins Z, Phillipou A, Rossell SL, et al. Assessing severity in anorexia nervosa: do the DSM-5 and an alternative severity rating based on overvaluation of weight and shape severity differ in psychological and biological correlates? Eur Eat Disord Rev. 2023;31(4):447–61.
- Machado BC, Gonçalves SF, Martins C, Hoek HW, Machado PP. Risk factors and antecedent life events in the development of anorexia nervosa: a Portuguese case-control study. Eur Eat Disord Rev. 2014;22(4):243–51.
- Stice E, Van Ryzin MJ. A prospective test of the temporal sequencing of risk factor emergence in the dual pathway model of eating disorders. J Abnorm Psychol. 2019;128(2):119.
- Bulik CM, Slof-Op't Landt MC, van Furth EF, Sullivan PF. The genetics of anorexia nervosa. Annu Rev Nutr. 2007;27:263–75.
- Solmi M, Radua J, Stubbs B, Ricca V, Moretti D, Busatta D, et al. Risk factors for eating disorders: an umbrella review of published meta-analyses. Braz J Psychiatry. 2020;43:314–23.
- Schellings G, Van Hout-Wolters B. Measuring strategy use with self-report instruments: theoretical and empirical considerations. Metacogn Learn. 2011;6:83–90.
- 11. Fairbun CG. Risk factors for bulimia nervosa. A community based casecontrol study. Arch Gen Psychiatry. 1997;54:509–17.
- Fairburn CG, Cooper Z, Doll HA, Welch SL. Risk factors for anorexia nervosa: three integrated case-control comparisons. Arch Gen Psychiatry. 1999;56(5):468–76.
- Pike K, Hilbert A, Wilfley DE, Fairburn C, Dohm F-A, Walsh B, et al. Toward an understanding of risk factors for anorexia nervosa: a case-control study. Psychol Med. 2008;38(10):1443–53.
- Karwautz A, Rabe-Hesketh S, Hu X, Zhao J, Sham P, Collier D, et al. Individual-specific risk factors for anorexia nervosa: a pilot study using a discordant sister-pair design. Psychol Med. 2001;31(2):317–29.
- Karwautz A, Wagner G, Waldherr K, Nader I, Fernandez-Aranda F, Estivill X, et al. Gene–environment interaction in anorexia nervosa: relevance of non-shared environment and the serotonin transporter gene. Mol Psychiatry. 2011;16(6):590–2.
- 16. Sun L, Sun Z, Wu L, Zhu Z, Zhang F, Shang Z, et al. Prevalence and risk factors for acute posttraumatic stress disorder during the COVID-19 outbreak. J Affect Disord. 2021;283:123–9.
- Arboix A. Cardiovascular risk factors for acute stroke: risk profiles in the different subtypes of ischemic stroke. World J Clin Cases. 2015;3(5):418.
- Machado PP, Grilo CM, Crosby RD. Evaluation of the DSM-5 severity indicator for anorexia nervosa. Eur Eat Disord Rev. 2017;25(3):221–3.
- Smith KE, Ellison JM, Crosby RD, Engel SG, Mitchell JE, Crow SJ, et al. The validity of DSM-5 severity specifiers for anorexia nervosa, bulimia nervosa, and binge-eating disorder. Int J Eat Disord. 2017;50(9):1109–13.
- Zayas LV, Wang SB, Coniglio K, Becker K, Murray HB, Klosterman E, et al. Gender differences in eating disorder psychopathology across DSM-5 severity categories of anorexia nervosa and bulimia nervosa. Int J Eat Disord. 2018;51(9):1098–102.
- Dalle Grave R, Sartirana M, El Ghoch M, Calugi S. DSM-5 severity specifiers for anorexia nervosa and treatment outcomes in adult females. Eat Behav. 2018;31:18–23.
- Gianini L, Roberto CA, Attia E, Walsh BT, Thomas JJ, Eddy KT, et al. Mild, moderate, meaningful? Examining the psychological and functioning correlates of DSM-5 eating disorder severity specifiers. Int J Eat Disord. 2017;50(8):906–16.
- Hay PJ, Touyz S, Sud R. Treatment for severe and enduring anorexia nervosa: a review. Aust N Z J Psychiatry. 2012;46(12):1136–44.
- Krug I, Binh Dang A, Granero R, Agüera Z, Sánchez I, Riesco N, et al. Drive for thinness provides an alternative, more meaningful, severity indicator than the DSM-5 severity indices for eating disorders. Eur Eat Disord Rev. 2021;29(3):482–98.
- Abbate-Daga G, Pierò A, Gramaglia C, Gandione M, Fassino S. An attempt to understand the paradox of anorexia nervosa without drive for thinness. Psychiatry Res. 2007;149(1–3):215–21.
- Penas-Lledo E, Fernandez-Aranda F, Jimenez-Murcia S, Granero R, Penelo E, Soto A, et al. Subtyping eating disordered patients along drive for thinness and depression. Behav Res Ther. 2009;47(6):513–9.

- Giles S, Hughes EK, Fuller-Tyszkiewicz M, Treasure J, Fernandez-Aranda F, Karwautz AF, et al. Bridging of childhood obsessive-compulsive personality disorder traits and adult eating disorder symptoms: a network analysis approach. Eur Eat Disord Rev. 2022;30(2):110–23.
- Nakai Y, Nin K, Noma S, Teramukai S, Fujikawa K, Wonderlich SA. The impact of DSM-5 on the diagnosis and severity indicator of eating disorders in a treatment-seeking sample. Int J Eat Disord. 2017;50(11):1247–54.
- Keller MB, Lavori PW, Friedman B, Nielsen E, Endicott J, McDonald-Scott P, et al. The longitudinal interval follow-up evaluation: a comprehensive method for assessing outcome in prospective longitudinal studies. Arch Gen Psychiatry. 1987;44(6):540–8.
- Warshaw MG, Dyck I, Allsworth J, Stout RL, Keller MB. Maintaining reliability in a long-term psychiatric study: an ongoing inter-rater reliability monitoring program using the longitudinal interval follow-up evaluation. J Psychiatr Res. 2001;35(5):297–305.
- Anderluh MB, Tchanturia K, Rabe-Hesketh S, Treasure J. Childhood obsessive-compulsive personality traits in adult women with eating disorders: defining a broader eating disorder phenotype. Am J Psychiatry. 2003;160(2):242–7.
- 32. Fairburn CG, Doll HA, Welch SL, Hay PJ, Davies BA, O'Connor ME. Risk factors for binge eating disorder: a community-based, case-control study. Arch Gen Psychiatry. 1998;55(5):425–32.
- Garner DM, Olmsted MP, Bohr Y, Garfinkel PE. The eating attitudes test: psychometric features and clinical correlates. Psychol Med. 1982;12(4):871–8.
- Garner DM, Olmsted MP, Polivy J, Garfinkel PE. Comparison between weight-preoccupied women and anorexia nervosa. Psychosom Med. 1984;46(3):255–66.
- Cohen J. Statistical power analysis for the behavioral sciences. Academic Press; 2013.
- Jafari M, Ansari-Pour N. Why, when and how to adjust your P values? Cell Journal (Yakhteh). 2019;20(4):604.
- Chen H, Cohen P, Chen S. How big is a big odds ratio? Interpreting the magnitudes of odds ratios in epidemiological studies. Commun Stat Simul Comput. 2010;39(4):860–4.
- Pike KM, So M, Hilbert A, Maekawa H, Shimanouchi T, Wilfley D, et al. Risk factors for anorexia nervosa and bulimia nervosa in Japan and compared to a US sample. Int J Eat Disord. 2021;54(2):155–67.
- Barakat S, McLean SA, Bryant E, Le A, Marks P, Touyz S, et al. Risk factors for eating disorders: findings from a rapid review. J Eat Disord. 2023;11(1):8.
- 40. Dahlenburg SC, Gleaves DH, Hutchinson AD. Anorexia nervosa and perfectionism: a meta-analysis. Int J Eat Disord. 2019;52(3):219–29.
- Lie SØ, RØØ, Bang L. Is bullying and teasing associated with eating disorders? A systematic review and meta-analysis. Int J Eat Disord. 2019;52(5):497–514.
- 42. Ålgars M. Shapes and sizes: body image, body dissatisfaction and disordered eating in relation to gender and gender identity. 2012.
- 43. Laraia BA, Leung CW, Tomiyama AJ, Ritchie LD, Crawford PB, Epel ES. Drive for thinness in adolescents predicts greater adult BMI in the growth and health study cohort over 20 years. Obesity. 2021;29(12):2126–33.
- Keski-Rahkonen A, Bulik CM, Neale BM, Rose RJ, Rissanen A, Kaprio J. Body dissatisfaction and drive for thinness in young adult twins. Int J Eat Disord. 2005;37(3):188–99.
- Dakanalis A, Timko CA, Colmegna F, Riva G, Clerici M. Evaluation of the DSM-5 severity ratings for anorexia nervosa in a clinical sample. Psychitry Res. 2018;262:124–8.
- Halmi KA, Sunday SR, Strober M, Kaplan A, Woodside DB, Fichter M, et al. Perfectionism in anorexia nervosa: variation by clinical subtype, obsessionality, and pathological eating behavior. Am J Psychiatry. 2000;157(11):1799–805.
- Melero S, Morales A, Espada JP, Fernández-Martínez I, Orgilés M. How does perfectionism influence the development of psychological strengths and difficulties in children? Int J Environ Res Public Health. 2020;17(11):4081.
- McGrath DS, Sherry SB, Stewart SH, Mushquash AR, Allen SL, Nealis LJ, et al. Reciprocal relations between self-critical perfectionism and depressive symptoms: evidence from a short-term, four-wave longitudinal study. Can J Behav Sci. 2012;44(3):169.
- 49. Matthews A, Kramer RA, Mitan L. Eating disorder severity and psychological morbidity in adolescents with anorexia nervosa or atypical anorexia

nervosa and premorbid overweight/obesity. Eat Weight Disord Stud Anorexia Bulimia Obes. 2022;8:1–10.

- Meierer K, Hudon A, Sznajder M, Leduc M-F, Taddeo D, Jamoulle O, et al. Anorexia nervosa in adolescents: evolution of weight history and impact of excess premorbid weight. Eur J Pediatr. 2019;178:213–9.
- Treasure J, Cardi V, Leppanen J, Turton R. New treatment approaches for severe and enduring eating disorders. Physiol Behav. 2015;152:456–65.
- Dakanalis A, Clerici M, Stice E. Prevention of eating disorders: Current evidence-base for dissonance-based programmes and future directions. Eat Weight Disord Stud Anorexia Bulimia Obes. 2019;24:597–603.
- 53. Tchanturia K, Hambrook D. Cognitive remediation therapy for anorexia nervosa. Treat Eat Disord Clin Handb. 2010;7:130.
- Dingemans AE, Danner UN, Donker JM, Aardoom JJ, Van Meer F, Tobias K, et al. The effectiveness of cognitive remediation therapy in patients with a severe or enduring eating disorder: a randomized controlled trial. Psychother Psychosom. 2013;83(1):29–36.
- Hay P, Chinn D, Forbes D, Madden S, Newton R, Sugenor L, et al. Royal Australian and New Zealand College of Psychiatrists clinical practice guidelines for the treatment of eating disorders. Aust N Z J Psychiatry. 2014;48(11):977–1008.
- Gadsby S. Explaining body size beliefs in anorexia. Cogn Neuropsychiatry. 2017;22(6):495–507.
- 57. Marzola E, Martini M, Brustolin A, Abbate-Daga G. Inpatients with severeenduring anorexia nervosa: understanding the "enduringness" specifier. Eur Psychiatry. 2021;64(1):e44.
- Fernández-Aranda F, Treasure J, Paslakis G, Agüera Z, Gimenez M, Granero R, et al. The impact of duration of illness on treatment nonresponse and drop-out: exploring the relevance of enduring eating disorder concept. Eur Eat Disord Rev. 2021;29(3):499–513.
- Glasofer DR, Muratore AF, Attia E, Wu P, Wang Y, Minkoff H, et al. Predictors of illness course and health maintenance following inpatient treatment among patients with anorexia nervosa. J Eat Disord. 2020;8:1–10.
- Takakura S, Aso CS, Toda K, Hata T, Yamashita M, Sudo N. Physical and psychological aspects of anorexia nervosa based on duration of illness: a cross-sectional study. BioPsychoSocial Med. 2019;13:1–7.
- Mora-Maltas B, Lucas I, Granero R, Vintró-Alcaraz C, Miranda-Olivos R, Baenas I, et al. Cognitive flexibility and DSM-5 severity criteria for eating disorders: assessing drive for thinness and duration of illness as alternative severity variables. J Eat Disord. 2023;11(1):155.
- 62. Murray SB, Nagata JM, Griffiths S, Calzo JP, Brown TA, Mitchison D, et al. The enigma of male eating disorders: a critical review and synthesis. Clin Psychol Rev. 2017;57:1–11.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

